**Author’s response to reviews**

**Title:** Feasibility of alcohol screening and brief intervention in primary health care in Kazakhstan: study protocol of a pilot cluster randomized trial

**Authors:**

Bernd Schulte (b.schulte@uke.de)

Amy O’Donnell (amy.odonnell@newcastle.ac.uk)

Harald Lahusen (harald.lahusen@uni-hamburg.de)

Christina Lindemann (ch.lindemann@uke.de)

Mariya Prilutskaya (mariyapril2407@gmail.com)

Oleg Yussopov (oyusopov@gmail.com)

Zhanar Kaliyeva (kalieva_zhanar@mail.ru)

Marcus Martens (marcus.martens@uni-hamburg.de)

Uwe Verthein (u.verthein@uke.uni-hamburg.de)

**Version:** 1  **Date:** 18 Nov 2019

**Author’s response to reviews:**

Dear editor, dear reviewers, many thanks for your fruitful comments. We have revised the manuscript accordingly as much as possible. You will find the specific responses in the attached table. Many thanks for your support, B. Schulte